Appendix 3G

Notification of issue, conversion or payment up of equity +securities

Information or documents not available now must be given to ASX as soon as available. Information and documents given to ASX become ASX's property and may be made public.

If you are an entity incorporated outside Australia and you are issuing a new class of +securities other than CDIs, you will need to obtain and provide an International Securities Identification Number (ISIN) for that class. Further information on the requirement for the notification of an ISIN is available from the Create Online Forms page. ASX is unable to create the new ISIN for non-Australian issuers.

*Denotes minimum information required for first lodgement of this form, with exceptions provided in specific notes for certain questions. The balance of the information, where applicable, must be provided as soon as reasonably practicable by the entity.

Part 1 – Entity and announcement details

Question no	Question	Answer
1.1	*Name of entity We (the entity here named) give notice of the issue, conversion or payment up of the following unquoted +securities.	Bravura Solutions Limited
1.2	*Registration type and number Please supply your ABN, ARSN, ARBN, ACN or another registration type and number (if you supply another registration type, please specify both the type of registration and the registration number).	ACN 164 391 128
1.3	*ASX issuer code	BVS
1.4	*This announcement is <i>Tick whichever is applicable.</i>	 A new announcement An update/amendment to a previous announcement A cancellation of a previous announcement
1.4a	*Reason for update Mandatory only if "Update" ticked in Q1.4 above. A reason must be provided for an update.	N/A
1.4b	*Date of previous announcement to this update Mandatory only if "Update" ticked in Q1.4 above.	N/A
1.4c	*Reason for cancellation Mandatory only if "Cancellation" ticked in Q1.4 above.	N/A
1.4d	*Date of previous announcement to this cancellation Mandatory only if "Cancellation" ticked in Q1.4 above.	N/A
1.5	*Date of this announcement	16 December 2020

Part 2 – Type of issue

Question No.	Question	Answer
2.1	*The +securities the subject of this notification are: Select whichever item is applicable. If you wish to notify ASX of different types of issues of securities, please complete a separate Appendix 3G for each type of issue.	 +Securities issued as a result of options being exercised or other +convertible +securities being converted and that are not to be quoted on ASX Partly paid +securities that have been fully paid up and that are not to be quoted on ASX +Securities issued under an +employee incentive scheme that are not being immediately quoted on ASX Other [please specify] If you have selected 'other' please provide the circumstances of the issue here:
2.2a.1	Please state the number and type of options that were exercised or other +convertible securities that were converted (including their ASX security code if available)? Answer this question if your response to Q2.1 is "securities issued as a result of options being exercised or other convertible securities being converted and that are not to be quoted on ASX".	N/A
2.2a.2	And the date the options were exercised or other +convertible securities were converted: Answer this question if your response to Q2.1 is "securities issued as a result of options being exercised or other convertible securities being converted and that are not to be quoted on ASX". Note: If this occurred over a range of dates, enter the date the last of the options was exercised or convertible securities was converted.	N/A
2.2b.1	Please state the number and type of partly paid +securities that were fully paid up (including their ASX security code if available)? Answer this question if your response to Q2.1 is "partly paid securities that have been paid up and that are not to be quoted on ASX".	N/A
2.2b.2	And the date the +securities were fully paid up: Answer this question if your response to Q2.1 is "partly paid securities that have been paid up and that are not to be quoted on ASX". Note: If this occurred over a range of dates, enter the date the last of the securities was fully paid up.	N/A

2.2c.1	Please state the number and +securities (including their A code) issued under an +emp incentive scheme that are no immediately quoted on ASX Answer this question if your respon "securities issued under an employ scheme that are not being immedia ASX".	SX security bloyee bt being se to Q2.1 is ee incentive	vest in August	formance rights which may 2023 (depending on of vesting conditions) BVSAA	
2.2c.2	*Please attach a document of details of a URL link for a do with ASX detailing the terms +employee incentive scheme summary of the terms. Answer this question if your respon "securities issued under an employed scheme that are not being immedia ASX".	of the e or a se to Q2.1 is ee incentive		tached a copy of the Plan gravura Solutions Limited entive Plan.	
2.2c.3	*Are any of these +securities to +key management person an +associate Answer this question if your respon "securities issued under an employe scheme that are not being immedia ASX".	nnel (KMP) or se to Q2.1 is ee incentive	Yes		
2.2c.3.a	*Provide details of the recipients and the number of +securities issued to each of them Answer this question if your response to Q2.1 is "securities issued under an employee incentive scheme tha not being immediately quoted on ASX" and your response to Q2.2c.3 is "Yes". Repeat the detail in the table below for each KMP involved in the issue. If the securities are being issued to the KMP, repeat the name of KMP or insert "Same" in "Name of registered holder". If the securities are being issued to an associate of a P insert the name of the associate in "Name of registered holder".		t are the		
	Name of KMP	Name of regist	ered holder	Number of +securities	
	Tony Klim	Tony Klim		224,824 performance rights	
	Martin Deda	Martin Deda		81,566 performance rights	
2.2d.1	*The purpose(s) for which th issuing the +securities is: Answer this question if your respon "Other". You may select one or more of the	se to Q2.1 is	 To fund the To pay for [provide de To pay for [provide de 	dditional working capital e retirement of debt the acquisition of an asset etails below] services rendered etails below] vide details below] ails:	

-		
2.2d.2	Please provide any further information needed to understand the circumstances in which you are notifying the issue of these +securities to ASX, including (if applicable) why the issue of the +securities has not been previously announced to the market in an Appendix 3B You must answer this question if your response to Q2.1 is "Other". If there is no other information to provide, please answer "Not applicable" or "N/A".	N/A
2.3	*The +securities being issued are: <i>Tick whichever is applicable</i>	 Additional +securities in an existing unquoted class that is already recorded by ASX ("existing class") New +securities in an unquoted class that is not yet recorded by ASX ("new class")

Part 3A - number and type of +securities being issued (existing class)

Question No.	Question	Answer
3A.1	*ASX security code & description	Performance rights which may vest in August 2023 (depending on achievement of vesting conditions)
		Security code: BVSAA
3A.2	*Number of +securities being issued	1,023,638
3A.3a	*Will the +securities being issued rank equally in all respects from their issue date with the existing issued +securities in that class?	Yes
3A.3b	*Is the actual date from which the +securities will rank equally (non-ranking end date) known? Answer this question if your response to Q3A.3a is	N/A
24.25	"No".	
3A.3c	*Provide the actual non-ranking end date Answer this question if your response to Q3A.3a is "No" and your response to Q3A.3b is "Yes".	N/A
3A.3d	*Provide the estimated non-ranking end period Answer this question if your response to Q3A.3a is "No" and your response to Q3A.3b is "No".	N/A
3A.3e	*Please state the extent to which the +securities do not rank equally:	N/A
	 in relation to the next dividend, distribution or interest payment; or 	
	• for any other reason Answer this question if your response to Q3A.3a is "No".	
	For example, the securities may not rank at all, or may rank proportionately based on the percentage of the period in question they have been on issue, for the next dividend, distribution or interest payment; or they may not be entitled to participate in some other event, such as an entitlement issue.	

Answer the questions in this part if your response to Q2.3 is "existing class".

Part 3B – number and type of +securities being issued (new class)

Answer the questions in this part if your response to Q2.3 is "new class".

Question No.	Question	Answer
3B.1	*Security description	
3B.2	*Security type Select one item from the list that best describes the securities the subject of this form. This will determine more detailed questions to be asked about the security later in this section. Select "ordinary fully or partly paid shares/units" for stapled securities or CDIs. For interest rate securities, please select the appropriate choice from either "Convertible debt securities" or "Non-	 Ordinary fully or partly paid shares/units Options +Convertible debt securities Non-convertible +debt securities Redeemable preference shares/units
	convertible debt securities". Select "Other" for performance shares/units and performance options/rights or if the selections available in the list do not appropriately describe the security being issued.	□ Other
3B.3	ISIN code Answer this question if you are an entity incorporated outside Australia and you are issuing a new class of securities other than CDIs. See also the note at the top of this form.	
3B.4	*Number of +securities being issued	
3B.5a	*Will all the +securities issued in this class rank equally in all respects from the issue date?	Yes or No
3B.5b	*Is the actual date from which the +securities will rank equally (non-ranking end date) known? Answer this question if your response to Q3B.5a is "No".	Yes or No
3B.5c	*Provide the actual non-ranking end date Answer this question if your response to Q3B.5a is "No" and your response to Q3B.5b is "Yes".	
3B.5d	*Provide the estimated non-ranking end period Answer this question if your response to Q3B.5a is "No" and your response to Q3B.5b is "No".	
3B.5e	*Please state the extent to which the +securities do not rank equally:	
	 in relation to the next dividend, distribution or interest payment; or 	
	for any other reason	
	Answer this question if your response to Q3B.5a is "No".	
	For example, the securities may not rank at all, or may rank proportionately based on the percentage of the period in question they have been on issue, for the next dividend, distribution or interest payment; or they may not be entitled to participate in some other event, such as an entitlement issue.	

		payment up of equity +securities
3B.6	Please attach a document or provide a URL link for a document lodged with ASX setting out the material terms of the +securities being issued You may cross reference a disclosure document, PDS, information memorandum, investor presentation or other announcement with this information provided it has been released to the ASX Market Announcements Platform.	
3B.7	*Have you received confirmation from ASX that the terms of the +securities are appropriate and equitable under listing rule 6.1? Answer this question only if you are an ASX Listing. (ASX Foreign Exempt Listings and ASX Debt Listings do not have to answer this question). If your response is "No" and the securities have any unusual terms, you should approach ASX as soon as possible for confirmation under listing rule 6.1 that the terms are appropriate and equitable.	Yes or No
3B.8a	Ordinary fully or partly paid shares/units Answer the questions in this section if you selected this	
	*+Security currency This is the currency in which the face amount of an issue is denominated. It will also typically be the currency in which distributions are declared.	
	*Will there be CDIs issued over the +securities?	Yes or No
	*CDI ratio Answer this question if you answered "Yes" to the previous question. This is the ratio at which CDIs can be transmuted into the underlying security (e.g. 4:1 means 4 CDIs represent 1 underlying security whereas 1:4 means 1 CDI represents 4 underlying securities).	X:Y
	*Is it a partly paid class of +security?	Yes or No
	*Paid up amount: unpaid amount Answer this question if answered "Yes" to the previous question. The paid up amount represents the amount of application money and/or calls which have been paid on any security considered 'partly paid'	X:Y
	The unpaid amount represents the unpaid or yet to be called amount on any security considered 'partly paid'. The amounts should be provided per the security currency (e.g. if the security currency is AUD, then the	
	paid up and unpaid amount per security in AUD).	
	*Is it a stapled +security? This is a security class that comprises a number of ordinary shares and/or ordinary units issued by separate entities that are stapled together for the purposes of trading.	Yes or No
3B.8b	Option details Answer the questions in this section if you selected this	security type in your response to Question 3B 2
	*+Security currency This is the currency in which the exercise price is payable.	security type in your response to Question 36.2.

7		
	*Exercise price	
	The price at which each option can be exercised and convert into the underlying security. If there is no exercise price please answer as \$0.00.	
	The exercise price should be provided per the security currency (i.e. if the security currency is AUD, the exercise price should be expressed in AUD).	
	*Expiry date	
	The date on which the options expire or terminate.	
	*Details of the number and type of +security (including its ASX security code if the +security is quoted on or recorded by ASX) that will be issued if an option is exercised For example, if the option can be exercised to receive one fully paid ordinary share with ASX security code ABC, please insert "One fully paid ordinary share (ASX:ABC)".	
3B.8c	Details of non-convertible +debt securities redeemable preference shares/units	s, +convertible debt securities, or
	Answer the questions in this section if you selected one	of these security types in your response to Question
	3B.2. Refer to Guidance Note 34 and the " <u>Guide to the Namin</u> <u>Debt and Hybrid Securities</u> " for further information on ce	ng Conventions and Security Descriptions for ASX Quoted ertain terms used in this section
	*Type of +security	□ Simple corporate bond
	Select one item from the list	□ Non-convertible note or bond
		Convertible note or bond
		Preference share/unit
		Capital note
		□ Hybrid security
		□ Other
	*+Security currency	
	This is the currency in which the face value of the security is denominated. It will also typically be the currency in which interest or distributions are paid.	
	Face value	
	This is the principal amount of each security.	
	The face value should be provided per the security currency (i.e. if security currency is AUD, then the face value per security in AUD).	
	*Interest rate type	□ Fixed rate
	Select one item from the list	Floating rate
	Select the appropriate interest rate type per the terms of the security. Definitions for each type are provided	Indexed rate
	in the Guide to the Naming Conventions and Security Descriptions for ASX Quoted Debt and Hybrid	Variable rate
	Securities	Zero coupon/no interest
		□ Other
	Frequency of coupon/interest payments	Monthly
	per year Select one item from the list.	Quarterly
		Semi-annual
		Annual
		No coupon/interest payments
		□ Other

First interest payment date A response is not required if you have selected "No coupon/interest payments" in response to the question above on the frequency of coupon/interest payments Interest rate per annum % p.a. Answer this question if the interest rate type is fixed. *Is the interest rate per annum estimated at this time? Yes or No Answer this question if the interest rate type is fixed. Yes or No If the interest rate per annum is estimated, then what is the date for this information to be announced to the market (if known) Answer this question if the interest rate type is fixed and your response to the previous question is "Yes". Answer "Unknown" if the date is not known at this time. *Does the interest rate include a reference rate, base rate or market rate (e.g. BBSW or CPI)? Yes or No *What is the reference rate, base rate or Yes or No
coupon/interest payments" in response to the question above on the frequency of coupon/interest payments Interest rate per annum Answer this question if the interest rate type is fixed. % p.a. *Is the interest rate per annum estimated at this time? Yes or No Answer this question if the interest rate type is fixed. Yes or No If the interest rate per annum estimated, then what is the date for this information to be announced to the market (if known) Answer this question if the interest rate type is fixed and your response to the previous question is "Yes". Answer "Unknown" if the date is not known at this time. *Does the interest rate include a reference rate, base rate or market rate (e.g. BBSW or CPI)? Yes or No *What is the reference rate, base rate or Yes or No
Answer this question if the interest rate type is fixed. Yes or No *Is the interest rate per annum estimated at this time? Yes or No Answer this question if the interest rate type is fixed. Yes or No If the interest rate per annum is estimated, then what is the date for this information to be announced to the market (if known) Yes or No Answer this question if the interest rate type is fixed and your response to the previous question is "Yes". Answer "Unknown" if the date is not known at this time. *Does the interest rate include a reference rate, base rate or market rate (e.g. BBSW or CPI)? Yes or No Answer this question if the interest rate type is floating or indexed Yes or No
*Is the interest rate per annum estimated at this time? Yes or No Answer this question if the interest rate type is fixed. If the interest rate per annum is estimated, then what is the date for this information to be announced to the market (if known) Answer this question if the interest rate type is fixed and your response to the previous question is "Yes". Answer "Unknown" if the date is not known at this time. *Does the interest rate (e.g. BBSW or CPI)? Yes or No *What is the reference rate, base rate or *What is the reference rate, base rate or Yes or No
at this time? Answer this question if the interest rate type is fixed. If the interest rate per annum is estimated, then what is the date for this information to be announced to the market (if known) Answer this question if the interest rate type is fixed and your response to the previous question is "Yes". Answer "Unknown" if the date is not known at this time. *Does the interest rate include a reference rate, base rate or market rate (e.g. BBSW or CPI)? Answer this question if the interest rate type is floating or indexed *What is the reference rate, base rate or
If the interest rate per annum is estimated, then what is the date for this information to be announced to the market (if known) Answer this question if the interest rate type is fixed and your response to the previous question is "Yes". Answer "Unknown" if the date is not known at this time. *Does the interest rate include a reference rate, base rate or market rate (e.g. BBSW or CPI)? Answer this question if the interest rate type is floating or indexed *What is the reference rate, base rate or
then what is the date for this information to be announced to the market (if known) Answer this question if the interest rate type is fixed and your response to the previous question is "Yes". Answer "Unknown" if the date is not known at this time. *Does the interest rate include a reference rate, base rate or market rate (e.g. BBSW or CPI)? Answer this question if the interest rate type is floating or indexed *What is the reference rate, base rate or
and your response to the previous question is "Yes". Answer "Unknown" if the date is not known at this time. *Does the interest rate include a reference rate, base rate or market rate (e.g. BBSW or CPI)? Answer this question if the interest rate type is floating or indexed *What is the reference rate, base rate or
time. *Does the interest rate include a reference rate, base rate or market rate (e.g. BBSW or CPI)? Yes or No Answer this question if the interest rate type is floating or indexed *What is the reference rate, base rate or
rate, base rate or market rate (e.g. BBSW or CPI)? Answer this question if the interest rate type is floating or indexed *What is the reference rate, base rate or
or indexed *What is the reference rate, base rate or
market rate?
Answer this question if the interest rate type is floating or indexed and your response to the previous question is "Yes".
*Does the interest rate include a margin above the reference rate, base rate or market rate?
Answer this question if the interest rate type is floating or indexed.
*What is the margin above the reference % p.a. rate, base rate or market rate (expressed as a percent per annum) Answer this question if the interest rate type is floating or indexed and your response to the previous question is "Yes".
*S128F of the Income Tax Assessment Act 🛛 s128F exempt
status applicable to the +security
Select one item from the list
For financial products which are likely to give rise to a payment to which s128F of the Income Tax S128F exemption status unknown S128F exemption status S128F exemption status
 "s128F exempt" means interest payments are not taxable to non-residents;
 "Not s128F exempt" means interest payments are taxable to non-residents;
 "s128F exemption status unknown" means the issuer is unable to advise the status;
 "Not applicable" means s128E is not applicable to
"Not applicable" means s128F is not applicable to this security
this security *Is the +security perpetual (i.e. no maturity Yes or No

This appendix is *not* available as an online form Please fill in and submit as a PDF announcement

*Select other features applicable to the +security Up to 4 features can be selected. Further information is available in the Guide to the Naming Conventions and Security Descriptions for ASX Quoted Debt and Hybrid Securities.	 Simple Subordinated Secured Converting Convertible Transformable Exchangeable Cumulative Non-Cumulative Redeemable Extendable Extendable Step-Down Step-Up Stapled
*Is there a first trigger date on which a right of conversion, redemption, call or put can be exercised (whichever is first)?	None of the aboveYes or No
*If yes, what is the first trigger date Answer this question if your response to the previous question is "Yes".	
Details of the number and type of +security (including its ASX security code if the +security is quoted on ASX) that will be issued if the securities to be quoted are converted, transformed or exchanged Answer this question if the security features include "converting", "convertible", "transformable" or "exchangeable". For example, if the security can be converted into 1,000 fully paid ordinary shares with ASX security code ABC, please insert "1,000 fully paid ordinary shares (ASX:ABC)".	

Part 4 – Issue details

Question No.	Question	Answer
4.1	*Have the +securities been issued yet?	Yes
4.1a	*What was their date of issue? Answer this question if your response to Q4.1 is "Yes".	16 December 2020
4.1b	*What is their proposed date of issue? Answer this question if your response to Q4.1 is "No".	N/A
4.2	*Are the +securities being issued for a cash consideration? If the securities are being issued for nil cash consideration, answer this question "No".	No

4.2a	*In what currency is the cash consideration being paid	N/A
	For example, if the consideration is being paid in Australian Dollars, state AUD.	
	Answer this question if your response to Q4.2 is "Yes".	
4.2b	*What is the issue price per +security	N/A
	Answer this question if your response to Q4.2 is "Yes" and by reference to the issue currency provided in your response to Q4.2a.	
	Note: you cannot enter a nil amount here. If the securities are being issued for nil cash consideration, answer Q4.2 as "No" and complete Q4.2c.	
4.2c	Please describe the consideration being provided for the +securities Answer this question if your response to Q4.2 is "No".	Performance rights are issued under the Bravura Solutions Limited Employee Incentive Plan for nil consideration.
4.3	Any other information the entity wishes to provide about the issue	N/A

Part 5 - Unquoted +securities on issue

Following the issue of the +securities the subject of this application, the unquoted issued +securities of the entity will comprise:

Note: the figures provided in the table in section 5.1 below are used to calculate part of the total market capitalisation of the entity published by ASX from time to time. Please make sure you include in the table each class of unquoted securities issued by the entity.

Restricted securities should be included in table 5.1.

5.1		* T (.)
	*ASX security code and description	*Total number of +securities on issue
	Fully paid ordinary shares (BVS)	247,165,023
	Performance rights (BVSAA): - which may vest in August 2021 (depending on achievement of vesting conditions)	801,352
	- which may vest in August 2022 (depending on achievement of vesting conditions)	659,433
	- which may vest in August 2023 (depending on achievement of vesting conditions)	1,023,638

Part 6 – Other Listing Rule requirements

The questions in this Part should only be answered if you are an ASX Listing (ASX Foreign Exempt Listings and ASX Debt Listings do not need to complete this Part) and:

- your response to Q2.1 is "+securities issued under an +employee incentive scheme that are not being immediately quoted on ASX"; or
- your response to Q2.1 is "Other"

Question No.	Question	Answer
6.1	*Are the securities being issued under Listing Rule 7.2 exception 13 ¹ and therefore the issue does not need any security holder approval under Listing Rule 7.1? Answer this question if your response to Q2.1 is "securities issued under an employee incentive scheme that are not being immediately quoted on ASX".	Yes
6.2	*Has the entity obtained, or is it obtaining, +security holder approval for the issue under listing rule 7.1? Answer this question if the response to Q6.1 is "No".	Yes
6.2a	*Date of meeting or proposed meeting to approve the issue under listing rule 7.1 Answer this question if the response to Q6.1 is "No" and the response to Q6.2 is "Yes".	24 November 2020
6.2b	*Are any of the +securities being issued without +security holder approval using the entity's 15% placement capacity under listing rule 7.1? Answer this question if the response to Q6.1 is "No" and the response to Q6.2 is "No".	N/A

¹ Exception 13

An issue of securities under an employee incentive scheme if within 3 years before the issue date:

- (a) in the case of a scheme established before the entity was listed a summary of the terms of the scheme and the maximum number of equity securities proposed to be issued under the scheme were set out in the prospectus, PDS or information memorandum lodged with ASX under rule 1.1 condition 3; or
- (b) the holders of the entity's ordinary securities have approved the issue of equity securities under the scheme as an exception to this rule. The notice of meeting must have included:
 - a summary of the terms of the scheme.
 - the number of securities issued under the scheme since the entity was listed or the date of the last approval under this rule;
 - the maximum number of +equity securities proposed to be issued under the scheme following the approval; and
 - a voting exclusion statement.

Exception 13 is only available if and to the extent that the number of +equity securities issued under the scheme does not exceed the maximum number set out in the entity's prospectus, PDS or information memorandum (in the case of (a) above) or in the notice of meeting (in the case of (b) above).

Exception 13 ceases to be available if there is a material change to the terms of the scheme from those set out in the entity's prospectus, PDS or information memorandum (in the case of (a) above) or in the notice of meeting (in the case of (b) above).

6.2b.1	*How many +securities are being issued without +security holder approval using the entity's 15% placement capacity under listing rule 7.1?	N/A
	Answer this question if the response to Q6.1 is "No", the response to Q6.2 is "No" and the response to Q6.2b is "Yes".	
	Please complete and separately send by email to your ASX listings adviser a work sheet in the form of Annexure B to Guidance Note 21 confirming the entity has the available capacity under listing rule 7.1 to issue that number of securities.	
6.2c	*Are any of the +securities being issued without +security holder approval using the entity's additional 10% placement capacity under listing rule 7.1A (if applicable)? <i>Answer this question if the response to Q6.1 is "No"</i> <i>and the response to Q6.2 is "No"</i> .	N/A
6.2c.1	*How many +securities are being issued without +security holder approval using the entity's additional 10% placement capacity under listing rule 7.1A? <i>Answer this question if the response to Q6.1 is "No",</i>	N/A
	the response to Q6.2 is "No" and the response to Q6.2c is "Yes".	
	Please complete and separately send by email to your ASX listings adviser a work sheet in the form of Annexure C to Guidance Note 21 confirming the entity has the available capacity under listing rule 7.1A to issue that number of securities.	

Introduced 01/12/19; amended 31/01/20

EMPLOYEE INCENTIVE PLAN RULES





Contents

1.	D	Definition and Interpretation	6
	1.1	Definitions	6
	1.2	Interpretation	9
	1.3	Application of Corporations Act and ASX Listing Rules	10
2.	In	nvitations	10
	2.1	Board may make Invitations	10
	2.2	Form of an Invitation	10
	2.3	Terms of grant	11
3.	C	Contribution arrangements	11
	3.1	Board may permit Contributions	11
	3.2	Invitations involving Contributions	11
	3.3	Default Contribution arrangements	11
4.	Lo	oan arrangements	12
	4.1	Lender may provide a Loan	12
	4.2	Invitations involving Loans	12
	4.3	Direction to pay the Loan	12
	4.4	Default Loan arrangements	12
5.	Т	rust arrangements	13
	5.1	Board may require a Trust	13
	5.2	Invitations involving a Trust	13
	5.3	Default Trust arrangements	13
6.	D	Dividend Equivalent Rights	14
	6.1	Board may grant	14
	6.2	Invitation	14
	6.3	Default rights	14
7.	Α	pplications	14
	7.1	How an Application is made	14
	7.2	Board discretions regarding Applications	14
	7.3	Consequences of an Application	15



8.	G	rant of Awards	15
	8.1	When grant occurs	15
	8.2	Grant in favour of the Eligible Employee	15
9.	Ve	esting and Exercise	15
	9.1	Vesting of an Option or a Performance Right	15
	9.2	Exercise of an Option	15
	9.3	Vesting of Restricted Shares	15
	9.4	Vesting of an Incentive Right	16
1().	Lapsing and forfeiture	16
	10.1	Lapsing of an Option	16
	10.2	Lapsing of a Performance Right	16
	10.3	Forfeiture of a Restricted Share	16
	10.4	Lapsing of an Incentive Right	16
	10.5	Forfeited Shares	17
1:	L.	Satisfaction of Options and Rights	17
	11.1	Allocation of Shares	17
	11.2	Cash equivalent payments	17
12	2.	Satisfaction of Incentive Rights	17
	12.1	Payment	17
13	3.	Satisfaction of Dividend Equivalent Rights	18
	13.1	Payment	
14	1.	Dealing Restrictions	18
	14.1	Limited Dealing in Awards	
	14.2	Consequence of prohibited Dealing	
	14.3	Allocated Shares	
	14.4	Enforcement of Dealing Restrictions	
1	5.	Cessation of Employment	18
	15.1	Treatment of an Award	
	15.2	Applicable treatment	
16	5.	Change of Employment	19
	16.1	Transfers outside Australia	



16.	2	Treatment of Awards	19
17.	Cor	ntrol of the Company	20
17.	1	Board determination	20
17.	2	Consequence of determination	20
17.	3	Exercise or Vesting following Control	20
18.	Adj	justments of Awards	20
18.	1	Adjustments circumstances	20
18.	2	Permitted adjustments	21
18.	3	Rights issues and entitlement offers	21
18.	4	Bonus issues	21
18.	5	Reorganisations	21
18.	6	Additional Awards or Dividend Equivalent Rights	21
18.	7	Notice to Participants	21
18.	8	Accumulation of adjustments	21
19.	Ina	ppropriate benefits	21
19.	1	Inappropriate circumstances	22
19.	2	Treatment of Awards	22
20.	Тах	٢	22
20.	1	Participant responsibility	22
20.	2	Withholding circumstances	22
20.	3	Reimbursement arrangements	23
21.	Sha	ares issued under the Plan	23
21.	1	Share ranking	23
21.	2	Listing of Shares on ASX	23
22.	Am	endments to the Plan	23
22.	1	Board may make amendments	23
22.	2	Restrictions on amendments	23
22.	3	Permitted amendments	23
22.	4	Termination or Suspension of the Plan	24
23.	Rel	ationship with Participants	24
23.	1	Separate rights and obligations	24



23.2	No compensation	24
24. Adr	ninistration of the Plan	24
24.1	Power of the Board	24
24.2	Waivers	24
24.3	Interpretation of the Plan	25
24.4	Appointment of attorney	25
24.5	Participant data	25
24.6	Overseas participants	25
24.7	Payments	25
24.8	Connection with other plans	25
24.9	Amounts owing by a Participant	25
25. Not	tices	26
25.1	Notices by the Company	26
25.2	Notices to a Group Company	26
25.3	Time of service	26
26. Gov	verning Law	26



Plan Rules

1. Definition and Interpretation

1.1 Definitions

After Tax Dividends means in respect of a Participant, the amount calculated in accordance with the following formula:

D - [(D + IC) x TR - IC]

where:

D = the aggregate dividend declared or determined to be paid on the Loan Shares Allocated to the Participant under the Plan and on any bonus shares allotted in respect of those Loan Shares.

IC = the imputation credit arising from the franking (if any) of D.

TR = the highest personal income tax rate under the Income Tax Rates Act 1986 (Cth) plus the Medicare levy under the Medicare Levy Act 1986 (Cth) plus any temporary budget repair levy under the Income Tax (Transitional Provisions) Act 1997 (Cth) plus any other levy which increases an individual's personal income tax liability but exclusive of any Medicare levy surcharge, as applicable in the relevant tax year.

Allocate means:

- a) the issue of a Share to or for the benefit of;
- b) procuring the transfer of a Share (pursuant to a purchase on-market or an off-market transfer) to or for the benefit of; or
- c) procuring the setting aside of a Share for the benefit of,

a Participant.

Application means an application by an Eligible Employee to participate in the Plan and be granted an Award.

ASIC means Australian Securities and Investment Commission.

ASX means ASX Limited ACN 008 624 691 or the financial market operated by it, as the context requires.

Award means:

- a) an Option;
- b) a Performance Right;
- c) a Restricted Share; or
- d) an Incentive Right.

Board means the board of directors of the Company, any committee of that board or a duly authorised person or body to which that board has delegated its powers under this Plan.

Cessation Date means in relation to a particular Participant, the date on which Cessation of Employment of that Participant occurs.

Cessation of Employment means in relation to a particular Participant:

- a) that Participant ceases to be an Employee for whatever reason; or
- b) that Participant's employer (being a company other than the Company) ceases to be a Group Company.

Change of Control Event means:

a) a Takeover Bid being made for Shares in the Company;



- b) a court orders a meeting to be convened in relation to a compromise or arrangement for the purposes of, or in connection with:
 - i. a scheme which would, if it becomes effective, result in any person (either alone or together with its related bodies corporate) owning all of the shares in the Company; or
 - ii. a scheme for the reconstruction of the Company or its amalgamation with any other company or companies;
 - iii. any other transaction or event that in the Board's opinion is likely to result in a change of Control of the Company.

Company means Bravura Solutions Limited ACN 164 391 128.

Constitution means the constitution of the Company

Contributions means a monetary amount to be applied towards the price payable for the Allocation of a Share, whether made before or after the Allocation, from one or more of the following:

- a) gross (before-tax) wages or salary;
- b) net (after-tax) wages or salary;
- c) a bonus or other remuneration.

Control has the meaning given in section 50AA of the Corporations Act.

Corporations Act means Corporations Act 2001 (Cth).

Dealing means in relation to an Award or a Share (as the case may be), any dealing, including but not limited to:

- a) a sale, transfer, assignment, trust, encumbrance, option, swap, any alienation of all or any part of the rights attaching to the Award or Share;
- b) any attempt to do any of the actions set out in paragraph (a); and
- c) any arrangement or transaction in financial products that operates to limit the economic risk associated with holding a Share or Award.

Dealing Restriction means the restrictions on Dealing with an Award or Shares (as applicable) as set out in these Rules or an Invitation.

Dividend Equivalent Right means a right to be paid a cash amount that is ultimately determined by reference to (wholly or in part):

- a) the amount or value of dividends paid or payable in relation to Shares over a specified period of time; or
- b) a change in the amount or value over a specified period of time of dividends paid or payable in relation to Shares.

Eligible Employee means:

- a) an Employee (including a director employed in an executive capacity); or
- b) any other person who the Board determines is eligible to receive a grant of an Award under the Plan.

Employee means any person who is in full-time or part-time employment of a Group Company.

Exercise Price means the amount payable to exercise an Option following Vesting as set out in an Invitation (as adjusted or amended in accordance with these Rules).

Financial Year means the financial year of the Company (being at the time of adoption of these Rules a period of 12 months starting on 1 July in one year and ending on 30 June in the following year).

Forfeited Share means a Share forfeited in accordance with these Rules.

Group means the Company and each Group Company.

Group Company means the Company and each of its Subsidiaries.

Incentive Right means a right to be paid a cash amount that is ultimately determined by reference to (wholly or in part):

a) the price or value of Shares at a specified point in time or period of time; or



b) a change in the price or value of Shares over a specified period of time.

Invitation means an invitation to an Eligible Employee made by the Board under clause 2.1 to apply to participate in the Plan and apply for a grant of an Award.

Lender means the Group Company which makes a Loan to a Participant pursuant to the terms of these Rules and an Invitation.

Loan means a loan by a Group Company to a Participant of all or part of the amount payable for the Allocation of Restricted Shares pursuant to the terms of the Rules and an Invitation.

Loan Balance means in relation to a Loan, the outstanding amount of the Loan, including any interest payable on the Loan, but less any amount forgiven on the Loan, pursuant to the terms of the Loan.

Loan Shares means Restricted Shares Allocated to a Participant pursuant to the Plan in respect of which a Loan was made to the Participant.

Listing Rules means the official listing rules of the ASX as they apply to the Company from time to time.

Market Value in relation to a Share means:

- a) the VWAP of Shares over the 10 Trading Days on which sales were recorded on ASX immediately before the relevant date; or
- b) any other calculation as determined by the Board.

Option means an option to acquire one Share (or to be paid a cash payment in lieu of one Share) in accordance with these Rules and an Invitation.

Participant means a person who has holds an Award or Share under the terms of this Plan from time to time.

Performance Right means a right to acquire one Share (or to be paid a cash payment in lieu of one Share) in accordance with these Rules and an Invitation.

Plan means the Bravura Solutions Limited Employee Incentive Plan as set out in these Rules.

Restricted Share means a Share Allocated under this Plan (other than as result of the exercise of an Option or the Vesting of a Performance Right).

Restriction Period means the period during which a Participant must not Deal with Awards or Shares Allocated on exercise or Vesting of an Award as specified in an Invitation.

Rules means the terms and conditions set out in this document as amended from time to time.

Securities Trading Policy means the policy of the Company on Dealing in securities of the Company as amended or replaced from time to time.

Share means a fully paid ordinary share in the capital of the Company.

Subsidiary means a body corporate which is a subsidiary of the Company within the meaning of section 9 of the Corporations Act.

Takeover Bid has the meaning given in section 9 of the Corporations Act.

Tax means any tax, levy, excise, duty, charge, surcharge, contribution, withholding tax, impost or withholding obligation of whatever nature, whether direct or indirect, by whatever method collected or recovered, together with any fees, penalties, fines, interest or statutory charge.

Tax Liability means a liability of the Participant for income tax or employment taxes under any wage, withholding or other arrangements or for any other tax, social security contributions or levy or charge of a similar nature.

Trading Day means any day on which the ASX is open for trading.

Trust means in relation to an Invitation, the trust specified by the Company as the trust for the purposes of the Invitation.



Trust Deed means the trust deed of the Trust, as amended from time to time

Trust Share means in respect of a Participant, a Share Allocated under the Plan that is held by the Trustee on behalf of that Participant pursuant to the Trust Deed.

Trustee means the trustee under the Trust Deed.

Unvested in respect of an Award means an Award which has not Vested.

Vest or Vesting means the process by which the holder of an Award becomes entitled to:

- a) in the case of an Option, exercise the Option in accordance with clause 9.2;
- b) in the case of a Performance Right, be Allocated a Share in accordance with clause 11.1; or
- c) in the case of an Incentive Right, be paid the amount payable in respect of the Incentive Right,

following all Vesting Conditions that apply to that Award being satisfied or otherwise waived by the Board.

Vesting Condition means one or more conditions which must be satisfied or circumstances which must exist before an Award Vests under these Rules.

Vesting Period means the period or periods over which the Vesting Conditions are measured or tested as specified by the Board for the purposes of a particular Award, including any period or periods over which any re-testing of Vesting Conditions occurs.

VWAP in relation to a Share mean the arithmetic average of the daily volume weighted average market price of the Shares (calculated to 2 decimal places of one cent) of all Shares sold on ASX excluding:

- a) special crossings, overseas trades, trades pursuant to the exercise of options, overnight trades, and any other trades that are excluded from an ASX volume weighted average price calculation; and
- b) any trades that the Board determines to exclude on the basis that the trades are not fairly reflective of natural supply and demand.

1.2 Interpretation

Headings are for convenience only and do not affect the interpretation of these Rules. The following rules of interpretation apply to these Rules unless the context requires otherwise:

- (a) a word importing the singular includes the plural and vice versa, and a word of any gender includes the corresponding words of any other gender;
- (b) if a word or phrase is given a defined meaning in this Plan, any other part of speech or grammatical form of that word or phrase has a corresponding meaning;
- (c) the word including or any other form of that word is not a word of limitation;
- (d) a reference to a statute includes any regulations or other instruments made under it (delegated legislation) and a reference to a statute or delegated legislation or a provision of either includes consolidations, amendments, reenactments and replacements;
- (e) a reference to a clause or schedule is to a clause of or a schedule to these Rules and a reference to these Rules includes any schedule to these Rules;
- (f) a reference to a Share includes a Restricted Share; and
- (g) an issue is not precluded from being a pro rata issue because certain holders of Shares are excluded from the issue or are allowed to subscribe for a greater number of securities than the entitlement of those holders, pursuant to the Corporations Act or Listing Rules.



1.3 Application of Corporations Act and ASX Listing Rules

- (a) Notwithstanding any other provision of these Rules, no Award may be offered under the Plan if to do so would contravene the Corporations Act, the Listing Rules or instruments of relief issued by ASIC from time to time relating to employee incentive schemes which the Company is relying on.
- (b) Notwithstanding any other provision of these Rules, Awards and Shares must not be issued, acquired, transferred or otherwise dealt with under the Plan if to do so would contravene the Corporations Act or the ASX Listing Rules.

2. Invitations

2.1 Board may make Invitations

- a) The Board may, from time to time, in its discretion invite Eligible Employees to participate in the Plan and apply for a grant of Awards upon the terms set out in the Plan and any other terms as the Board determines.
- b) The Board may in its discretion invite an Eligible Employee to apply for a grant of a Dividend Equivalent Right in an Invitation to apply for a grant of Award (other than a Restricted Share).
- c) Without limiting clause 2.1(a), an Invitation may be expressed as an offer to the Eligible Employee to participate in the Plan and may specify the manner in which the Eligible Employee may accept that offer or the conduct of the Eligible Employee which will be treated as acceptance of that offer, including the failure of the Eligible to lodge or submit an election not to participate in the Plan in accordance with the instructions that accompany the Invitation.
- d) Without limiting clause 2.1(a), an Invitation may be made to a person where the Invitation is conditional on the person becoming an Employee.

2.2 Form of an Invitation

d)

- a) An Invitation may take any form determined by the Board, including by electronic means or by way of making it available on a website and notifying the Eligible Employee recipient that it is available on the website.
- b) An Invitation must include the following information:
 - i. the type or types of Awards being offered;
 - ii. the number of Awards being offered or the method by which the number will be calculated.
 - iii. the amount (if any) that will be payable for the grant of an Award or the method by which that amount will be calculated; and
 - iv. any Vesting Conditions or other conditions that apply to the Award, including any Vesting Period.
- c) An Invitation for a grant of an Option or Performance Right must include the following information.
 - i. if the Award is an Option, the requirements for exercising the Option, including any Exercise Price that will be payable and the period or periods in which the Option may be exercised;
 - ii. whether the exercise of the Option or Vesting of the Performance Right will only be satisfied by an Allocation of Shares to the Participant; and
 - iii. the dates or circumstances in which Option or Performance Right may lapse.
 - An Invitation for a grant of an Incentive Right must include the following information:
 - i. how the amount of the Incentive Right will be calculated; and
 - ii. when the amount of the Incentive Right will be paid.
- e) An Invitation may include the following information:
 - i. the circumstances (if any) in which Shares Allocated to the Participant may be forfeited;
 - ii. any Dealing Restrictions (including the Restriction Period); and
 - iii. how an Award may be treated in the event that a Cessation of Employment of the Participant occurs;
 - iv. how an Award may be treated in the event that a Change of Control Event occurs;
 - v. how an Award may be treated in the circumstances referred to in clause 16.1; and
 - vi. how an Award may be treated in the circumstances referred to in clause 19.1.



2.3 Terms of grant

Unless an Invitation provides otherwise:

- (a) no payment will be required for the grant of an Award; and
- (b) an Option or Performance Right does not confer on a Participant the right to participate in new issue of securities to existing holders of Shares unless:
- (i) the Participant has become entitled to exercise the Option or the Performance Right has Vested;
- (ii) Shares have been Allocated under the Plan in respect of that Option or Performance Right before the record date for the determination of entitlements to the new issue of securities; and
- (iii) the Participant participates as a holder of those Allocated Shares.

3. Contribution arrangements

3.1 Board may permit Contributions

The Board may in its discretion determine that a Participant may satisfy the payment of the amount (if any) payable for the Allocation of Restricted Shares by means of the application of Contributions.

3.2 Invitations involving Contributions

If the Board makes a determination under clause 3.1, the Invitation for the grant of those Restricted Shares must specify:

- a) the form and manner by which a Participant can agree to make Contributions;
- b) the type of Contributions which may be made;
- c) the period or periods of time during which Contributions may be made;
- d) any limitations on the amount of Contributions which can be made, which may vary between different types of Contributions and may include either or both a minimum amount and maximum amount;
- e) the circumstances (if any) in which the amount of the Contributions agreed to be made by the Participant can be changed and how that change may be effected;
- f) the manner in which Contributions made by a Participant will be held pending their application for an Allocation of Shares;
- g) the procedure by which the Contributions will be applied towards the price payable for an Allocation of Shares, including the time or times when the Restricted Shares will be Allocated and how Contributions in excess of the price payable for the Allocation of the Restricted Shares will be dealt with;
- h) how the Contributions will be treated in the event a Cessation of Employment of the Participant occurs;
- i) how the Contributions will be treated in the event a Change of Control Event occurs;
- j) the circumstances (if any) in which the Participant may be prohibited or suspended from making further Contributions and the consequences of that prohibition or suspension, including how Contributions made prior to that time will be dealt with; and
- k) the procedure by which a Participant may elect to discontinue making Contributions under the Plan, when that election takes effect and the consequences of that election.

3.3 Default Contribution arrangements

Unless an Invitation to which this clause 3 applies provides otherwise:

 any Contributions (other than in the form of future gross (before-tax) salary or wages or from a Loan under the Plan) will pending their application be held by or on behalf of the Company or a Group Company on trust for Participants in an account with a financial institution that is used solely in connection with employee incentive schemes of the Group Companies(including the Plan);



- b) a Participant cannot elect to discontinue making Contributions in regard to Shares which have already been Allocated to the Participant;
- c) subject to clause 3.3(b), a notice given by the Participant to the Company in accordance with these Rules to discontinue making Contributions will take effect 45 days after the giving of the notice;
- d) if a Participant has elected to discontinue making Contributions in accordance with the procedure specified in the Invitation, any Contributions (other than in the form of future gross (before-tax) salary or wages or from a Loan under the Plan) that have not been applied for the Allocation of Shares to the Participant under the Plan will be repaid to the Participant as soon as practicable after that discontinuation less any applicable Tax on those Contributions; and
- e) the Company will not pay interest on Contributions made by a Participant pending the application of those Contributions for the Allocation of Shares to the Participant.

4. Loan arrangements

4.1 Lender may provide a Loan

The Board may in its discretion determine that a Loan may be made by the Lender to a Participant to be used to satisfy all or part of the payment of the amount payable for the grant of Restricted Shares.

4.2 Invitations involving Loans

If the Board makes a determination under clause 4.1, the Invitation for the grant of those Restricted Shares must specify:

- a) the amount of the Loan or the method by which that amount will be calculated;
- b) the identity of the Lender;
- c) the method by which an Eligible Employee may apply for the Loan or may accept an offer to be provided with a Loan;
- d) the dates on which or the circumstances in which the Participant must repay the Loan;
- e) the limitations (if any) on the Participant repaying the Loan before it is due and payable;
- f) whether interest is payable by the Participant on the Loan and, if so, how that interest is calculated;
- g) whether the Participant must give a mortgage, charge or other security interest to secure the obligation to repay the Loan;
- h) the circumstances (if any) in which the obligation to repay all or part of the Loan may be forgiven by the Lender;
- i) the recourse of the Lender against the Participant in relation to repayment of the Loan;
- j) how the Loan will be treated in the event of a Cessation of Employment of the Participant;
- k) the rights of the Company to Deal with Loan Shares if the Participant does not repay the Loan and how the proceeds will be applied; and
- I) how the Loan will be treated in the event that a Change of Control Event occurs.

4.3 Direction to pay the Loan

Each Participant to whom a Loan is made irrevocably directs the Lender to apply the amount of the Loan towards payment of the amount payable for Allocation of the Loan Shares of that Participant.

4.4 Default Loan arrangements

Unless the Invitation in respect of a Loan provides otherwise:

- a) no fees or interest will be payable by a Participant in respect of a Loan;
- b) subject to the recourse of the Lender against the Participant, until the Loan of a Participant is repaid in full, the Company has a lien over:
 - i. all of the Loan Shares held by that Participant to which the Loan relates;
 - ii. all dividends and other amounts paid or payable on those Loan Shares; and
 - iii. all Shares or other securities issued in respect of those Loan Shares as part of a bonus issue or entitlement offer;



- c) except as provided by clause 4.4(b), a Participant will not be required to give a mortgage, charge or other security interest in respect of the Loan;
 - the recourse of the Lender against the Participant in relation to repayment of the Loan Balance is limited to:
 - i. the proceeds of sale of the Loan Shares by the Company; and
 - ii. the forfeiture of the Loan Shares;

d)

- e) if any dividends are declared or determined to be paid by the Company on Loan Shares in respect of which there is a Loan Balance, then the Participant irrevocably authorises and directs the Company to pay or apply the After Tax Dividend on the Shares first towards the payment of interest (if any) on the Loan and then towards the repayment of the remaining Loan Balance;
- f) if any cash amount (other than dividends) is to be paid by the Company on Loan Shares in respect of which there is a Loan Balance, then the Company may and the Participant irrevocably authorises and directs the Company to pay or apply that cash amount first towards the payment of interest (if any) on the Loan and then towards the repayment of the remaining Loan Balance; and
- g) a Participant may not participate in any dividend reinvestment plan (or similar plan) established by the Company in respect of Loan Shares of the Participant until the Loan in respect of those Loan Shares has been fully repaid.

5. Trust arrangements

5.1 Board may require a Trust

The Board may in its discretion determine that a Restricted Share will be held by a Trustee on behalf of a Participant on the terms and conditions of the Trust Deed and any additional terms as the Board determines.

5.2 Invitations involving a Trust

If the Board make a determination under clause 5.1, the Invitation for those Restricted Shares must set out:

- a) the name of the Trust and Trustee;
- b) the rights of the Participant to receive income deriving from the Trust Shares, including dividends and distributions;
- c) the rights of the Participant to exercise, or to direct the Trustee to exercise on their behalf, any voting rights attaching to Trust Shares; and
- d) the circumstances in which the Participant, or the Company on behalf of the Participant, may direct the Trustee to:
 - i. transfer the Trust Share into the Participant's name; or
 - ii. sell the Trust Share and pay the proceeds of sale less any transaction costs of the sale to the Participant.

5.3 Default Trust arrangements

Unless the terms of the Trust Deed or an Invitation in respect of Trust Shares provides otherwise:

- a) the Trustee will not levy any fees or charges for administering the Trust that are payable directly by the Participant or out of the assets of the Trust, other than reasonable disbursements including brokerage and Tax levied or incurred in connection with the Trust;
- b) subject to any terms applying to Loan Shares, the Participant is entitled to receive all dividends and other distributions or benefits payable to the Participant or the Trustee in respect of Trust Shares;
- c) the Participant is entitled to direct the Trustee in writing how to exercise the voting rights attaching to Trust Shares;
- d) where the Trustee is a Group Company, the Trustee will not, at its own discretion, exercise the voting rights attaching to Trust Shares;
- e) any bonus shares that are issued in respect of Trust Shares will be issued to the Trustee on the Participant's behalf and will be held subject to the same Dealing Restrictions as the Trust Shares in respect of which they were issued;
 f) if rights arise on a rights issue or entitlement offer in respect of Trust Shares:
 - if rights arise on a rights issue or entitlement offer in respect of Trust Shares:
 i. the Participant may instruct the Trustee how to deal with those rights in accordance with the Trust Deed; and



- ii. if the Participant does not instruct the Trustee how to deal with those rights, the rights will be dealt with in accordance with the Trust Deed; and
- g) where a Trust Share ceases to be a Restricted Share, the Trustee will continue to hold the Trust Share on trust on behalf of the Participant until the Participant, or the Company on behalf of the Participant, directs the Trustee to:
 - i. transfer the Trust Share into the Participant's name; or
 - ii. sell the Trust Share in accordance with the Trust Deed and pay the proceeds of sale less any transaction costs of the sale to the Participant.

6. Dividend Equivalent Rights

6.1 Board may grant

The Board may determine at the time an Invitation of Options, Performance Rights or Incentive Rights is made that the Participant will also be granted a Dividend Equivalent Right in respect of:

- a) Options which are exercised; or
- b) Performance Rights which Vest; or
- c) Incentive Rights which Vest.

6.2 Invitation

If the Board makes a determination under clause 6.1, the Invitation for the grant of Dividend Equivalent Rights must specify:

- a) how the amount of the Dividend Equivalent Right will be calculated; and
- b) when the amount of the Dividend Equivalent Right will be paid.

6.3 Default rights

Unless an Invitation of Dividend Equivalent Rights provides otherwise:

- a) the amount of the Dividend Equivalent Rights which apply to an Option or a Performance Right will be approximately equal to the amount of dividends that would have been payable to a Participant had he or she been the registered holder of the Shares which are Allocated on exercise of that Option or Vesting of that Performance Right from the first day of the Financial Year in which that Option or Performance Right is granted (excluding any dividends actually paid in respect of those Shares after their Allocation to the Participant);
- b) the amount of the Dividend Equivalent Rights will not be grossed up or otherwise adjusted to account for any Tax consequences which would have applied if the Participant had actually been paid a dividend; and
- c) the Company must satisfy the entitlement of a Participant to receive a payment of a Dividend Equivalent Right which apply to an Option or a Performance Right as soon as reasonably practicable following the Allocation of Shares on exercise of that Option or Vesting of that Performance Right.

7. Applications

7.1 How an Application is made

An Application by the Eligible Employee must be made in accordance with the instructions that accompany the Invitation to which the Application relates, or in any other way the Board determines.

7.2 Board discretions regarding Applications

a) The Board may, at its discretion, refuse to allow the participation of a person in the Plan where that person ceases to be an Eligible Employee, or ceases to satisfy any other condition imposed by the Board, before the grant of the Award is made.



b) Nothing limits the Board's ability to treat the conduct of an Eligible Employee in respect of an Invitation as valid application to participate in the Plan under these Rules.

7.3 Consequences of an Application

By applying to participate in the Plan in respect of an Invitation, the Eligible Employee is deemed to have agreed to be bound by these Rules, the terms of the Invitation and the Constitution.

8. Grant of Awards

8.1 When grant occurs

If:

- a) an Application by a Participant is accepted by the Board; or
- b) an Invitation in the form referred to in clause 2.1(c) is made and that offer is accepted in accordance with the terms of the Invitation,

the Board will (subject to its discretion under clause 7.2):

- c) in the case of an Award other than Restricted Shares, grant the Award to the Eligible Employee; or
- d) in the case of Restricted Shares, Allocate Shares to the Eligible Employee.

8.2 Grant in favour of the Eligible Employee

Unless the Board determines otherwise and subject to clause 5, an Award may not be registered in any name other than that of the Eligible Employee.

9. Vesting and Exercise

9.1 Vesting of an Option or a Performance Right

- a) Subject to any express clause of these Rules, an Option or a Performance Right will only Vest (and in the case of an Option become exercisable) where each Vesting Condition and any additional terms specified in the Invitation for that Option or Performance Right have been satisfied or otherwise waived by the Board.
- b) Vesting occurs upon notification from the Company to the Participant that an Option or Performance Right has Vested pursuant to this clause 9.1.

9.2 Exercise of an Option

- a) An Option may only be exercised if it has Vested.
- b) The exercise of an Option may only be effected in a form and manner specified in the Invitation or as otherwise determined by the Board.
- c) Subject to clause 11.2(b)(ii), on the exercise of an Option the Participant must pay the Exercise Price for that Option.

9.3 Vesting of Restricted Shares

Subject to any express clause of these Rules, a Share ceases to be a Restricted Share where:

- a) each Vesting Condition and any additional terms specified in the Invitation for that Restricted Share have been satisfied or otherwise waived by the Board; and
- b) the Company notifies the Participant that the Share is no longer a Restricted Share.



9.4 Vesting of an Incentive Right

- a) Subject to any express clause of these Rules, an Incentive Right will only Vest where each Vesting Condition and any additional terms specified in the Invitation for that Incentive Right have been satisfied or otherwise waived by the Board.
- b) Vesting occurs upon notification from the Company to the Participant that an Incentive Right has Vested pursuant to this clause 9.4.

10. Lapsing and forfeiture

10.1 Lapsing of an Option

An Option will lapse upon the earliest to occur of:

- a) 7 years after Vesting or any other date specified in the Invitation for that Option;
- b) a date or circumstance specified in the Invitation for that Option or a provision of these Rules as when an Option lapses;
- c) failure to meet a Vesting Condition within the Vesting Period or meet any other condition applicable to the Option within the period specified in the Invitation for that Option; or
- d) the receipt by the Company of a notice in writing from a Participant that the Participant has elected to surrender the Option.

10.2 Lapsing of a Performance Right

A Performance Right will lapse upon the earliest to occur of:

- a) a date or circumstance specified in the Invitation for that Performance Right or a provision of these Rules as when a Performance Right lapses;
- b) failure to meet a Vesting Condition within the Vesting Period or meet any other condition applicable to the Performance Right within the period specified in the Invitation for that Performance Right; or
- c) the receipt by the Company of a notice in writing from a Participant that the Participant has elected to surrender the Performance Right.

10.3 Forfeiture of a Restricted Share

A Restricted Share will be forfeited upon the earliest to occur of:

- a) a date or circumstance specified in the Invitation for that Restricted Share or a provision of these Rules as when a Restricted Share is forfeited;
- b) failure to meet a Vesting Condition within the Vesting Period or any other condition applicable to the Restricted Share within the period specified in the Invitation for that Restricted Share; or
- c) the receipt by the Company of a notice in writing from a Participant that the Participant has elected to surrender the Restricted Share.

10.4 Lapsing of an Incentive Right

An Incentive Right will lapse upon the earliest to occur of:

- a) a date or circumstance specified in the Invitation for that Incentive Right or a provision of these Rules as when an Inventive Right lapses;
- b) failure to meet a Vesting Condition within the Vesting Period or meet any other condition applicable to the Incentive Right within the period specified in the Invitation for that Incentive Right; or
- c) the receipt by the Company of a notice in writing from a Participant that the Participant has elected to surrender the Incentive Right.



10.5 Forfeited Shares

- a) Where a Forfeited Share is held by the Participant, the Participant is taken to have agreed to dispose of all of his or her interest in that Share for no consideration and the Share will be transferred into the name of the Company's nominee.
- b) Where a Forfeited Share is held by the Trustee, the Participant's rights in the Share will be extinguished for no consideration and the Share will be held by the Trustee as general trust property in accordance with the terms of the Trust.
- c) Where a Forfeited Share was Allocated to a Participant on the exercise of an Option, the Company may, but is not obliged to, repay to the Participant any Exercise Price paid by that Participant in relation to that Forfeited Share.

11. Satisfaction of Options and Rights

11.1 Allocation of Shares

Subject to clause 11.2, on the exercise of an Option or Vesting of a Performance Right, the Company must Allocate a Share to the Participant.

11.2 Cash equivalent payments

- a) Unless the terms of an Invitation provides otherwise, the Board may determine that the exercise of an Option or the Vesting of a Performance Right will be satisfied by the Company making a cash payment to the Participant in lieu of an Allocation of Shares. The Board may determine that some or all of a Participant's Options or Rights will be settled in this manner.
- b) Where the Board makes a determination under 11.2(a), the Company must notify the Participant of the following information:
 - i. the number of Options or Performance Rights the subject of that determination; and
 - ii. that no Exercise Price is payable in respect of an Option the subject of that determination or that the Company will refund any Exercise Price paid by the Participant in respect of an Option the subject of that determination.
- c) Where the Board makes a determination under 11.2(a), the Company must, as soon as reasonably practicable following the exercise of an Option or the Vesting of a Performance Right, pay to the Participant an amount equivalent to the value of the exercised Options or the Vested Performance Rights the subject of that determination calculated in accordance with clause 11.2(d).
- d) The amount of the cash payment referred in clause 11.2(c) will be calculated by multiplying the number of Options that have been exercised or the number of Performance Rights that have Vested (as applicable) by the Market Value on the date on which the Share in respect of that Option or Performance Right would otherwise have been Allocated to a Participant, less in the case of Options, any Exercise Price of those Options specified in the Invitation which has not been paid by the Participant to the Company.

12. Satisfaction of Incentive Rights

12.1 Payment

The Company must pay the amount of a Vested Incentive Right on the payment date specified in the Invitation for that Incentive Right.



13. Satisfaction of Dividend Equivalent Rights

13.1 Payment

The Company must pay the amount of a Dividend Equivalent Right on the payment date specified in the Invitation for that Dividend Equivalent Right.

14. Dealing Restrictions

14.1 Limited Dealing in Awards

A Participant must not Deal with an Award unless that Dealing:

- a) is required by law and the Participant has provided satisfactory evidence of that requirement to the Board; and
- b) is permitted (or is not prohibited) by the Company's Securities Trading Policy.

14.2 Consequence of prohibited Dealing

- a) Where, in the opinion of the Board, a Participant Deals with an Award (other than a Restricted Share) in breach of clause 14.1, the Award will immediately lapse.
- b) Where, in the opinion of the Board, a Participant Deals with his or her interest in a Restricted Share in breach of clause 14.1, the Restricted Share is deemed to immediately be forfeited.

14.3 Allocated Shares

The Board may, at its discretion, impose a restriction on Dealing with Shares Allocated on exercise of an Option or Vesting of a Performance Right.

14.4 Enforcement of Dealing Restrictions

- a) The Company may implement any procedure it considers appropriate to enforce a Dealing Restriction, including the imposition of a holding lock or requiring that Shares be held in trust on behalf of a Participant.
- b) By making an Application or accepting an Invitation, each Participant undertakes not to take any action or permit another person to take any action to remove any procedure imposed by the Company under clause 14.4(a).

15.Cessation of Employment

15.1 Treatment of an Award

- a) The Board may:
 - i. specify in the terms of an Invitation; or
 - ii. make a determination as to,

how a Participant's Awards will be treated on the occurrence of Cessation of Employment of the Participant.

- b) The terms of an Invitation may specify that:
 - i. that the applicable treatment under clause 15.2 is subject to the Board's power to make a determination under clause 15.1(a)(ii); or
 - ii. subject to applicable laws, the circumstances in which a Cessation of Employment of a Participant will be treated as not having occurred for the purposes of the Plan.

15.2 Applicable treatment

a) The applicable treatment under clause 15.1(a) may be that some or all of the Participant's Awards:



- i. Vest on the Cessation Date, or on the occurrence of a specified date or circumstance, or on the satisfaction of a specified condition;
- ii. in the case of an Option, is only exercisable within a specified period or on the satisfaction of a specified condition;
- iii. lapse or are forfeited; or
- iv. are not subject to a term of the grant of the applicable Award, including a Vesting Condition or a Disposal Restriction.
- b) In making a determination under clause 15.1(a), the Board may have regard to any matter the Board considers relevant, including:
 - i. the circumstances in which the Cessation of Employment of the Participant occurred;
 - ii. the extent that the Vesting Conditions of the Award are determined or estimated by the Board to have been satisfied when tested at the Cessation Date; and
 - iii. the proportion of the Vesting Period of the Award that has elapsed at the time of the Cessation Date.
- c) Notwithstanding any rule, no Group Company is required to pay, provide, or procure the payment or provision of, any money or other benefit to the Participant which would require shareholder approval under the Corporations Act, if such relevant approval has not been obtained. Nothing in the Plan requires or will be deemed to require any Group Company to seek the approval of their respective shareholders to enable them to perform an action in connection with an Award.

16. Change of Employment

16.1 Transfers outside Australia

The Board may:

- a) specify in the terms of an Invitation; or
- b) make a determination as to,

how a Participant's Awards will be treated where a Participant is transferred to work for a Group Company outside Australia and, as a result of that transfer:

- c) the Participant would suffer a tax disadvantage in relation to their Awards which is demonstrated to the satisfaction of the Board;
- d) the Company would be restricted in its ability to Vest Awards or to Allocate Shares upon the exercise of Options or the Vesting of Performance Shares;
- e) the Company would be restricted in its ability to pay the amount payable on Vesting of an Incentive Right; or
- f) the Participant would become subject to restrictions on their ability to Deal with the Awards, or to hold or Deal in the Shares or the proceeds of the Shares acquired on Vesting or exercise, because of the laws of the country to which the Participant is transferred.

16.2 Treatment of Awards

The applicable treatment under clause 16.1 may be that some or all of the Participant's Awards:

- a) Vest on the occurrence of a specified date or circumstance, or on the satisfaction of a specified condition;
- b) in the case of an Option, is exercisable within a specified period or on the satisfaction of a specified condition;
- c) an Option or Performance Right will be satisfied by the Company making a cash payment to the Participant in lieu of an Allocation of Shares; or
- d) any another manner determined by the Board or specified in an Invitation.



17. Control of the Company

17.1 Board determination

- a) Subject to clause 17.1(b), where a Change of Control Event occurs, the Board may, in its discretion, determine that:
 - i. all or a specified number of a Participant's Unvested Awards are deemed to have Vested;
 - ii. all or a specified number of a Participant's Options may be exercised for a period specified by the Board, and if not exercised within that period, will lapse;
 - iii. the Disposal Restrictions or any other terms which apply to the Award cease to apply;
 - iv. the Disposal Restrictions which apply to Shares Allocated on the Vesting of a Performance Right or exercise of an Option cease to apply; and/or
 - v. the Company on behalf of the Participant will direct the Trustee to transfer Trust Shares into the Participant's name.
- b) The terms of an Invitation may specify a particular treatment of a Participant's Unvested Awards where a Change of Control Event occurs, and the Invitation may specify that such treatment is subject to the Board's power to make a determination under clause 17.1(a).

17.2 Consequence of determination

- a) Where the Board makes a determination pursuant to clause 17.1(a), the Board must as soon as reasonably practicable give written notice to each Participant of the number of Awards that have Vested or may be exercised (as the case maybe).
- b) If the Board determines under clause 17.1(a) that only some of a Participant's Unvested Awards will vest, all remaining Unvested Awards will lapse, unless the Board determines a different treatment.

17.3 Exercise or Vesting following Control

If:

- a) a body corporate (Acquirer) obtains Control of the Company as a result of a Change of Control Event; and
- b) the Company, the Acquirer and the Participant agree,

a Participant may, subject to applicable laws, upon exercise of Options or Vesting of Performance Rights be provided with shares of the Acquirer or its parent in lieu of Shares:

- c) in such manner as the parties agree; and
- d) on substantially the same terms and subject to substantially the same conditions as the Shares but with appropriate adjustments to the number and kind of shares of the Acquirer.

18. Adjustments of Awards

18.1 Adjustments circumstances

Prior to:

- a) the Allocation of Shares upon the exercise of Options or the Vesting of Performance Rights or the making a cash payment in lieu of an Allocation of Shares to a Participant in accordance with clause 11.2; or
- b) the payment to a Participant of the amount payable on a Vested Incentive Right or payment of a Dividend Equivalent Right,

the Board may, in its discretion, determine in respect of Awards (other than Restricted Shares) or Dividend Equivalent Rights to:

c) make adjustments to the terms of an Award or Dividend Equivalent Right granted to that Participant; or



d) grant additional Awards or Dividend Equivalent Rights to that Participant,

in order to minimise or eliminate any material advantage or disadvantage to a Participant resulting from a corporate action by the Company or a capital reconstruction of the Company, including a return of capital.

18.2 Permitted adjustments

The applicable adjustment under clause 18.1 may be to one or more of the following:

- a) the number of Awards (other than Restricted Shares) or Dividend Equivalent Rights to which each Participant is entitled;
- b) the number of Shares to which each Participant is entitled upon exercise of Options or Vesting of Performance Rights;
- c) the amount payable upon Vesting of Incentive Rights;
- d) the amount payable upon exercise of Options (including the Exercise Price);
- e) the amount payable upon the Vesting of Performance Rights;
- f) the amount payable to a Participant under a Dividend Equivalent Right.

18.3 Rights issues and entitlement offers

Without limiting clause 18.1, if the Company makes a pro rata issue of Shares to existing holders of Shares generally by way of a rights issue or entitlement offer and no Share has been Allocated in respect of an Option before the record date for determining entitlements to the pro rata issue, the Option will be adjusted in accordance with the formula in the Listing Rules.

18.4 Bonus issues

Without limiting clause 18.1, if the Company makes a bonus issue of Shares to existing holders of Shares (other than an issue in lieu of dividends or by way of a dividend reinvestment) involving capitalisation of reserves of distributable profits and no Shares has been Allocated in respect of an Option or Performance Right before the record date for determining entitlements to the bonus issue, the Option or Performance Right will be adjusted in accordance with the Listing Rules.

18.5 Reorganisations

Without limiting clause 18.1, if a reorganisation (including consolidation, subdivision, reduction or return) of the issued capital of the Company is effected, Options and Performance Rights will be adjusted in accordance with the Listing Rules.

18.6 Additional Awards or Dividend Equivalent Rights

- a) Where additional Awards (other than Restricted Shares) are granted to the Participant under this clause 18, such Awards will be subject to the same terms and conditions as the original Awards granted to the Participant (including any Vesting Conditions) unless the Board determines otherwise.
- b) Where additional Dividend Equivalent Rights are granted to the Participant under this clause 18, such Dividend Equivalent Rights will be subject to the same terms and conditions as the original Dividend Equivalent Rights granted to the Participant unless the Board determines otherwise.

18.7 Notice to Participants

The Board must as soon as reasonably practicable after making any additional grants or adjustments under this clause 18, give notice in writing of the adjustment to any affected Participant.

18.8 Accumulation of adjustments

Effect will be given to this clause 18 in a manner that the effect of the successive application of this clause 18 will be cumulative, with the intention being that the adjustments that it progressively effects will reflect previous adjustments.

19. Inappropriate benefits



19.1 Inappropriate circumstances

The Board may make a determination how a Participant's Awards will be treated where, in the opinion of the Board:

- a) an Award of Participant Vests or may Vest as a result of the fraud, dishonesty or breach of duties or obligations of any person to any Group Company and the Award would not have otherwise Vested;
- b) a Participant:
 - i. has acted fraudulently or dishonestly;
- c) has breached his or her duties or obligations to any Group Company; or
 - i. has done an act which brings any Group Company into disrepute;
- d) there has been a material misstatement or omission in the financial statements of a Group Company or an event or circumstance has occurred which will require the financial statements of a Group Company to be restated; or
- e) the Company is required by or entitled under law or a policy of a Group Company to clawback remuneration of a Participant.

19.2 Treatment of Awards

- a) The applicable treatment under clause 19.1 may be that some or all of the Participant's:
 - i. Unvested Awards;
 - ii. Shares that have not ceased to be Restricted Shares under these Rules;
 - iii. Vested but unexercised Options;
 - iv. Shares allocated upon the exercise of Options or Vesting of Performance Rights;
 - v. Dividend Equivalent Rights;
 - vi. Shares that have ceased to be Restricted Shares under these Rules,

will lapse or be deemed to be forfeited (as the case maybe).

- b) The applicable treatment under clause 19.1 may be that the Participant must pay or repay (as the case maybe) to the Company as a debt due to the Company any of the following:
 - i. a cash payment received in lieu of an Allocation of Shares pursuant to clause 11.2;
 - ii. a cash payment made in respect of an Incentive Right;
 - iii. a cash payment made in respect of a Dividend Equivalent Right;
 - iv. all or part of the net proceeds of sale of Shares Allocated under the Plan which have been sold;
 - v. dividends received in respect of a Share Allocated under the Plan.
- c) The applicable treatment under clause 19.1 may be any other circumstance specified in an Invitation.

20.Tax

20.1 Participant responsibility

- a) Unless otherwise required by law, no Group Company is responsible for any Tax which may become payable by a Participant as consequence of or in connection with the Participant's participation in the Plan, including:
 - i. the grant, Vesting or exercise of an Award;
 - ii. the Allocation of Shares; or
 - iii. any Dealing with an Award or any Shares.
- b) No Group Company makes any representation that the Plan will have any particular Tax or financial consequences or that an Eligible Person or Participant will gain any Tax or financial advantage by participating in the Plan.

20.2 Withholding circumstances

If a Group Company is obliged, or reasonably believes it may have an obligation, as a result of or in connection with:

- a) the grant of an Award to a Participant, or the Vesting or exercise of any Award;
- b) the payment of any cash amount to a Participant, including upon the Vesting or exercise of any Award; or



c) the Allocation of Shares to, or on behalf of, a Participant under the Plan,

to account for a Tax Liability, then the Participant must reimburse the Group Company for the amount or amounts so paid or payable.

20.3 Reimbursement arrangements

If a Participant is obliged to reimburse a Group Company for a Tax Liability in accordance with clause 20.2, the Group Company is not obliged to grant the Awards, pay the relevant amount or Allocate the relevant Shares to the Participant unless the Group Company is satisfied that arrangements have been made for reimbursement of the Group Company.

21. Shares issued under the Plan

21.1 Share ranking

Any Shares issued under the Plan will rank equally in all respects with other Shares for the time being on issue by the Company except as regards any rights attaching to such Shares by reference to a record date prior to the date of their issue.

21.2 Listing of Shares on ASX

If Shares of the same class as those issued on the Vesting or exercise of an Award are quoted on ASX, the Company will apply for quotation of Shares issued under the Plan within the period required by the ASX Listing Rules.

22. Amendments to the Plan

22.1 Board may make amendments

- a) Subject to clause 22.2, the Board may at any time by resolution:
 - i. amend all or any of the provisions of the Plan; or
 - ii. amend the terms or conditions of any Award granted under the Plan.
- b) Notwithstanding clause 22.2, the Board may waive, amend or replace any Vesting Condition of an Award if the Board determines that the original Vesting Condition is no longer appropriate or applicable, provided that the interests of the relevant Participant are not, in the opinion of the Board, materially prejudiced or advantaged relative to the position reasonably anticipated at the time of the grant of the Award.
- c) Any exercise by the Board of a discretion contemplated by these Rules or the terms of an Invitation will not constitute an amendment pursuant to this clause 22.

22.2 Restrictions on amendments

Subject to clause 22.3, the Board may not exercise its powers under clause 22.1(a) in a manner which adversely affects the existing rights of the Participant in respect of any granted Award or Share already Allocated except with the consent of the Participant or with approval of at least 75% of the Participants holding Awards affected by an amendment.

22.3 Permitted amendments

Clause 22.2 does not apply to an amendment which the Board considers necessary or desirable to:

- a) comply with or to take account of a change in legislation, the Listing Rules, exchange control, or other regulatory requirement governing or regulating the maintenance or operation of the Plan or similar plans, in any jurisdiction in which invitations under the Plan have been made;
- b) correct any manifest error or mistake; or
- c) take into consideration possible adverse tax implications in respect of the Plan arising from adverse rulings, changes to tax legislation or changes in the interpretation of tax legislation by a court of competent jurisdiction.



22.4 Termination or Suspension of the Plan

- a) The Board may, at any time, terminate or suspend the Plan.
- b) The termination or suspension of the Plan will not affect any existing Awards granted under the Plan and the terms of the Plan will continue to apply to such Awards provided that, in the case of termination, all Shares Allocated under the Plan then subject to a Dealing Restriction will be released from the restriction on the date of termination or on such other date specified by the Board.

23. Relationship with Participants

23.1 Separate rights and obligations

- a) Unless the subject of an express provision in an employment contract, the rights and obligations of any Eligible Employee under the terms of their office, employment or contract with a Group Company are not affected by their participation in the Plan.
- b) These Rules do not form part of, and are not incorporated into, any contract of any Participant (whether or not they are an employee of a Group Company).
- c) Participation in the Plan does not confer on a Participant the right to remain an Employee and does not affect any rights which a Group Company may have to terminate the employment of a Participant.
- d) Participation in the Plan on a particular basis for the grant of an Award does not create any right or expectation of the grant of further Awards on the same basis, or at all.

23.2 No compensation

No Participant has any right to compensation for any loss in relation to the Plan, including:

- a) any loss or reduction of any rights or expectations under the Plan in any circumstances or for any reason (including lawful or unlawful termination of employment or the employment relationship);
- b) any exercise of a discretion or a decision taken in relation to a grant of Awards or in relation to the Plan, or any failure to exercise a discretion under these Rules;
- c) the operation, suspension, termination or amendments of the Plan; or
- d) lapse or forfeiture of an Award.

24. Administration of the Plan

24.1 Power of the Board

- a) The Plan will be administered by the Board.
- b) The Board has power to:
 - i. determine appropriate procedures for administration of the Plan consistent with these Rules;
 - ii. appoint or engage service providers for the operation and administration of the Plan; and
 - iii. delegate to any one or more persons for any period and on any conditions as it may determine the exercise of any of its powers or discretions arising under the Plan.
- c) Except as otherwise expressly provided in the Plan, the Board has absolute and unfettered discretion to act or refrain from acting under or in connection with the Plan and in the exercise of any power or discretion under the Plan.

24.2 Waivers

Notwithstanding any other provisions of the Plan, the Board may at any time waive in whole or in part any terms or conditions (including any Vesting Condition) in relation to any Awards granted to any Participant.



24.3 Interpretation of the Plan

In the event of any dispute, disagreement or uncertainty as to the interpretation of the Rules, or as to any question or right arising from or related to the Plan or to any Awards granted or Shares Allocated under it, the decision of the Board is final and binding.

24.4 Appointment of attorney

Each Participant appoints the company secretary of the Company (or any other officer of the Company authorised by the Board for this purpose) as his or her attorney to do anything necessary to:

- a) Allocate Shares to the Participant in accordance with these Rules;
- b) effect a forfeiture of Shares in accordance with these Rules or an Invitation; and
- c) execute transfers of Shares in accordance with these Rules.

24.5 Participant data

By participating in the Plan, each Participant consents to the holding and processing of personal data provided by the Participant to the Company for all purposes relating to the operation of the Plan. These include, but are not limited to:

- a) administering and maintaining Participants' records;
- b) providing information to trustees of any trust (including the Trustee), registrars, brokers or third party administrators of the Plan or to any regulatory authority where required by law;
- c) providing information to future purchasers of the Company or the business in which the Participant works; and
- d) transferring information about the Participant to a country or territory outside Australia.

24.6 Overseas participants

- a) The Board determine that additional rules apply to an Award that is granted to persons who are not a resident of Australia by attaching a schedule to these Rules.
- b) A determination under clause 24.6(a) may specify how the rules the subject of that determination interact with these Rules.
- c) Unless the determination under clause 24.6(a) provides otherwise, the rules the subject of the determination prevail over any other provision of these Rules to the extent of any inconsistency.

24.7 Payments

- a) The Company may make a payment of an amount payable pursuant to these Rules in Australian dollars or any other currency determined by the Board in its discretion. The Company may make payments in different currencies to different Participants. The Board may determine the appropriate exchange rate and time of calculation of the amount of a payment made in a currency other than Australian dollars.
- b) Where the Board determines that the payment under these Rules is to be made in a currency other than Australian dollars, unless the Board determines otherwise, the foreign exchange rate applied will be the average closing exchange rate listed on the Reserve Bank of Australia website of the relevant currency for the 5 days prior to the date of payment.

24.8 Connection with other plans

Participation in the Plan does not affect, and is not affected by, participation in any other incentive or other plan operated by the Group Companies unless the terms of that other plan provides otherwise or unless otherwise stated in the Invitation.

24.9 Amounts owing by a Participant

Where a Participant owes any amount or amounts to a Group Company, including any overpayment of leave or wages or salary, or any loss suffered by a Group Company as a result of any breach of contract, statutory duty or tort committed by the Participant, the Board may, in respect of any Awards granted to the Participant:



- a) prevent the exercise of some or all of the Awards;
- b) determine that some or all of the Awards lapse; or
- c) reduce the number of Awards which Vest,

to take into account of and in settlement of any such amounts.

25.Notices

25.1 Notices by the Company

Any notice, consent or other communication under or in connection with the Plan may be given by the Company to an Eligible Employee or Participant if:

- a) delivered personally to the addressee or sent by prepaid post to his or her last known residential address or to the address of the place of business at which the Participant performs all or most of their duties (Place of Business);
- b) sent to him or her by facsimile or email to his or her last notified fax number or email address or to fax number or email address at the Place of Business; or
- c) posted on any intranet or website maintained by the Company or an administrator of the Plan and accessible by that Eligible Employee or Participant.

25.2 Notices to a Group Company

Any notice, consent or other communication under or in connection with the Plan may be given by an Eligible Employee or Participant to a Group Company if delivered or by sending it by post or facsimile to its registered office (or any other address notified by that company from time to time for that purpose (Notified Address)) or the fax number (if any) of that registered office (or Notified Address).

25.3 Time of service

- a) Where a notice or other communication is given by post, it is deemed to have been received:
 - a. if sent to an address within Australian, 48 hours it is posted; or
 - b. if sent to an address outside Australia, 5 days after it was posted.
- b) Where a notice or other communication is given by facsimile or email, it is deemed to have been received on completion of transmission.

26. Governing Law

The Rules and the rights of Eligible Employees and Participants under the Plan are governed by and must be construed according to the law applying in New South Wales, Australia.